Author's response to reviews

Title: Prevalence and determinants of pre-hypertension and hypertension among the adults in rural Bangladesh: Findings from a community-based study

Authors:

Masuma A Khanam (masuma.khanam@uon.edu.au)
Wietze Lindeboom (wietze.lindeboom@gmail.com)
Abdur Razzaque (razzaque@icddrb.org)
Louis Niessen (lniessen@jhsph.edu)
Abul H Milton (milton.hasnat@newcastle.edu.au)

Version: 2 Date: 8 October 2014

Author's response to reviews: see over
To,
Editor-in-Chief,
BMC Public Health

Dear Editor,

Please find the revised manuscript titled “Prevalence and determinants of pre-hypertension and hypertension among the adults in rural Bangladesh: Findings from a community-based study”, submitted for publication in the Journal BMC Public Health.

We have now inserted line numbers, changed ‘Introduction’ to ‘Background’ and provided the list of abbreviation, as suggested.

Thank you for your consideration of our research work.

Sincerely,

Abul Hasnat Milton

Corresponding author